# **RESEARCH ARTICLE**

**Open Access** 

# Effects of autism on 30-year outcome of anorexia nervosa



Søren Nielsen<sup>1</sup>, Sandra Rydberg Dobrescu<sup>2</sup>, Lisa Dinkler<sup>2</sup>, Carina Gillberg<sup>2</sup>, Christopher Gillberg<sup>2,3</sup>, Maria Råstam<sup>2,4</sup> and Elisabet Wentz<sup>5,6\*</sup><sup>1</sup>

## Abstract

**Background:** Long-term consequences of comorbid autism spectrum disorder (ASD) in individuals with anorexia nervosa (AN) are inadequately investigated.

**Methods:** In the 1980s, 51 adolescent-onset AN cases (AN group) and 51 matched controls (COMP group) were recruited from the community. They have been examined on five occasions. The four last assessments included the Morgan-Russell Outcome Assessment Schedule (MROAS) to assess eating disorder outcomes (weight, dieting, menstruation), and related problems including psychiatric, psychosexual and socioeconomic state. In the present study, at age 44, when 30 years had elapsed, MROAS data were compared with previous results. At age 16, 21, 24 and 32 years, all individuals had been assessed regarding ASD. At the 30-year follow-up, the impact of the ASD on the MROAS data was analysed.

**Results:** In the AN group, all core anorectic symptoms (weight, dieting, menstruation) were on a par with the COMP group at the 30-year follow-up, but the positive outcomes were limited to those who had never had an ASD diagnosis assigned. The AN group—again particularly those with ASD—had a more negative attitude to sexual matters than the COMP group. The AN group had worse outcomes than the COMP group for 'personal contacts', 'social contacts', and 'employment record' at the 30-year follow-up and the outcomes were worse the more often an ASD diagnosis had been assigned.

**Limitations:** Rare data collection points throughout 30 years (only 5 assessments). ASD was assessed in the first four studies but was not assessed again at the 30-year follow-up.

**Conclusions:** Mental health, psychosexual, and socioeconomic status were compromised up to 30 years after AN onset. Coexisting ASD contributed to the poor outcome. Core anorectic symptoms had "normalised" three decades after AN onset.

**Plain English summary:** Some individuals with anorexia nervosa (AN) also suffer from autism. In this study we have investigated outcome of AN 30 years after the onset of AN and whether the presence of autism affects the outcome. Since the 1980s we have followed 51 individuals with teenage-onset AN and 51 healthy controls. They have been examined on five occasions, and an instrument that measures symptoms of AN (weight, dieting, body image), psychiatric symptoms, ability to work, and relationships with partner, family, and friends has been used to assess outcome.

Full list of author information is available at the end of the article



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

<sup>\*</sup>Correspondence: elisabet.wentz@gu.se

<sup>&</sup>lt;sup>5</sup> Department of Psychiatry and Neurochemistry, Institute

of Neuroscience and Physiology, University of Gothenburg, Gothenburg, Sweden

Autism was assessed in the first four studies. Symptoms of AN had normalised at 30-year follow-up, but only among those without autism. Psychiatric symptoms, ability to work, and relationships were issues that persisted after 30 years in the AN group, and those who had both autism and a history of AN had even more pronounced problems in these areas. The AN group had a more negative attitude to sexual matters than the control group, the outcome was worse the more often an autism diagnosis had been assigned.

**Conclusions:** Mental health, psychosexual, and socioeconomic status are affected up to 30 years after AN onset, particularly among those with autism.

**Keywords:** Anorexia nervosa, Autism spectrum disorder, Long-term outcome, Psychosexual, Socioeconomic, Community-based, Controlled

#### Background

Anorexia nervosa (AN) is characterised by starvation, underweight, fear of gaining weight, and a distorted body perception. It is one of the most severe psychiatric disorders that can affect a young person, due to a considerable risk of a chronic course and the highest mortality rate among all mental illnesses [1]. AN affects almost every vital organ, including the cerebral, cardiovascular, gastrointestinal, endocrinological, and musculoskeletal systems [2–4]. The increased mortality in AN is mainly caused by starvation, resulting in conditions such as treatmentrefractory infections, cardiac arrest and kidney failure [5]. Comorbid alcohol use disorder and suicide are other well recognised causes of death in individuals with AN [1, 6].

Systematic data regarding very long-term psychosexual and socioeconomic consequences of AN are rare [7-10]. Ratnasuriya's [7] and Löwe's groups [8] followed up the psychosexual and socioeconomic status of inpatients with AN after 20 and 21 years, respectively. Fifty to sixty per cent of the former inpatients were living with a partner, and 50–71% were employed. In a German long-term outcome study, more than a thousand AN inpatients were followed for an average of ten years, including a subsample with 20-year outcome data [11]. At an average age of 34.5, only 27.4% of the women had become mothers compared with 57-74% of women of a similar age in the German general population [11]. Ratnasuriya et al. [7], using the Morgan Russell Assessment Schedule (MROAS) [12], observed that after 20 years, at mean age 41 years, one in three lived a very isolated life, and one in four had failed to emancipate from their family of origin. Non-eating disorder (ED) psychiatric morbidity at very long-term follow-up is dominated by affective and anxiety disorders [8, 13]. According to Dobrescu et al., 37.8% had a non-ED psychiatric diagnosis 30 years after the onset of AN [13]. Very long-term outcome studies pertaining to Health-Related Quality of Life (HRQoL) have shown poorer mental status in individuals with a history of AN [13], and poorer mental and physical HRQoL in individuals with chronic AN [14].

Autism spectrum disorder (ASD) has been reported to be overrepresented in chronic cases of AN [15]. In the Gothenburg AN study, we found that ASD in childhood or in young adulthood predicted a poor outcome at the 18-year follow-up of adolescent-onset AN [9]. The Gothenburg AN study has followed prospectively a group of adolescent-onset AN cases recruited from the community in 1985 [16]. MROAS has been used as an outcome measure at all follow-up examinations, and ASD diagnoses have been assessed since the original study in 1985 [17]. Nielsen et al. [10] analysed MROAS data including the effect of ASD over an 18-year follow-up period. Mental health, and psychosexual and socioeconomic state were poorer in the AN group, and the outcome was worse if an ASD was present. No other studies have previously followed prospectively a group of adolescent-onset AN cases for 30 years, and used the well-established outcome instrument MROAS at each follow-up examination. Furthermore, no one has hitherto investigated whether an ASD diagnosis will affect the outcome according to the MROAS 30 years after the onset of AN.

The aim of the present study was to investigate the course of the core anorexic traits, mental health, psychosexual and socioeconomic state in our AN group over a 30-year period and compare the data with the matched control group. We aimed specifically to analyse the effect of the diagnostic stability of ASD on the scales and subscales of the MROAS. Based on discouraging results pertaining to the psychosocial and socioeconomic outcome in previous long-term follow-up studies, we hypothesised that our AN group would still exhibit poorer results at the 30-year follow-up than their matched controls regarding mental health and psychosexual and socioeconomic state. We further hypothesised that a previously assigned ASD diagnosis would have a negative impact on MROAS, in general, and on the domains assessing mental health and psychosexual and socioeconomic state, in particular.

## Methods

### Participants

In the mid 1980's an epidemiological study of the prevalence of adolescent-onset AN was conducted in Gothenburg, Sweden. All 4291 individuals born in 1970 and living in Gothenburg in 1985 were screened for previous and present AN. They all completed a questionnaire pertaining to ED symptoms and a researcher (Maria Råstam; MR) scrutinised all the growth charts. MR examined all individuals with a suspicion of an ED, resulting in 23 girls and two boys with AN. In-depth examination was declined by one of the girls leaving 22 girls and 2 boys to form the population-based group. Another group, the population-screening group, consisting of 27 individuals with adolescent-onset AN (26 girls, one boy) born in 1969, 1971-1975 and 1977, was formed after the cases had been reported to the researchers by the school health services. The *population-based* group and the population-screening group differed regarding ED treatment received, but virtually all other aspects were similar between the groups. The two groups were therefore merged to form the AN group consisting of 51 individuals; 48 girls and three boys (for further details see Råstam et al. 1989 and Råstam 1992) [16, 17]. All the AN cases fulfilled the criteria for AN according to the DSM-III-R [18] and the DSM-IV [19]. The mean age at onset of AN was 14.3 years. The school health services were also asked to select school-, age-, and sex matched comparison cases, without a history of ED. The comparison cases constituted the COMP group and included 51 individuals (48 girls and three boys) in line with the AN group format.

#### Procedure

The 51 AN and 51 COMP cases were examined thoroughly for the first time at mean age 16 years (AN Study 1). The assessment included a collateral interview with the mother. The 102 cases (51 AN and 51 COMP) have thereafter been prospectively followed up at four occasions, at mean age 21, 24, 32, and 44 years [9, 13, 20, 21]. The follow-up periods were approximately 6 (AN Study 2), 10 (AN Study 3), 18 (AN Study 4), and 30 years (AN Study 5) after onset of AN, respectively. There was no attrition in AN Study 2, AN Study 3, and AN Study 4. In AN Study 4, there were 45 face-to-face interviews and five telephone interviews in the AN group. In one case, a mother was interviewed instead of the daughter, as the daughter was currently suffering from severe AN. In the COMP group, 48 individuals were interviewed in person, and three through telephone interviews. In the latest follow-up, AN Study 5, all but two women and two men in the AN group, and all individuals in the COMP group agreed to participate, corresponding to a dropout rate of 4% for the whole sample. In *AN Study 5* online video conferences or telephone interviews were conducted with eleven and nine of the individuals in the AN and COMP group, respectively. All the other participants were interviewed face-to-face.

ASD diagnoses were assigned in AN Study 1, AN Study 2, AN Study 3, and AN Study 4, each time by a new rater blinded to group status. The ASD diagnoses were based on structured interviews in all four studies. In addition, in AN Study 2, the Dewey social awareness test [22], and in AN Study 4, the self-report Autism-Spectrum Quotient questionnaire [23], were used. For details on the instruments used to assess symptoms of ASD see Nielsen et al. [10]. In AN study 4 the individuals were grouped based on whether they had been assigned an ASD diagnosis in AN Study 1, AN Study 2, AN Study 3, and AN Study 4 at one or several occasions. Individuals in the AN group with an ASD diagnosis at all four examinations were categorized as "ASD  $\times$  4" (n=6), those who had been assigned an ASD diagnosis at least once and at most three times belonged to the "ASD  $\times$  1–3" category (n = 10), and the rest, those who had never fulfilled criteria for ASD, were classified as "never ASD" (n=34). One woman in the AN group had experienced a severe head trauma in her early 20's. She had previously been classified as having ASD, but in the present study she was excluded, since her psychiatric symptoms were not considered to have a childhood onset but acquired in early adult years. A woman in the COMP group was also removed from the study since her ASD diagnosis was considered secondary to a longstanding substance use disorder. The remaining 50 individuals in the COMP group were classified as "never ASD". The thorough investigations of ASD traits in AN Study 1 to AN Study 4 resulted in no further diagnostic assessments of ASD in AN Study 5.

#### Instruments

The semi-structured Mini International Neuropsychiatric Interview (MINI 6.0) [24] was used to interview all individuals regarding psychiatric disorders, both current disorders and disorders during the twelve-year period that had elapsed since the 18-year follow-up study (*AN Study 4*). The ED domain of the Structured Clinical Interview for DSM-IV (SCID-I) [25] was used as a complement to MINI 6.0, due to the additional questions regarding AN and binge-eating disorder. A checklist for DSM-5 EDs was included since the MINI and SCID-I interviews are based on DSM-IV criteria.

The Morgan-Russell outcome assessment schedule (MROAS) is a tool for estimating ED outcome, and was launched by Morgan & Russell in the 1970's [26] and modified by Morgan & Hayward approximately a decade later [12]. The instrument comprises five scales (A

to E), where scale A, D and E also include subscales. The questions focus on the previous six months. The scales/ subscales are scored from 0 to 12 points. Scale A is entitled "Food intake", with the subscales "A1. Dietary restriction" (0 corresponds to dietary restriction "at all times", and 12 "nil" dietary restriction; 5 response categories), "A2. Worry about body weight or appearance" (0: worry "at all times"; 12: "nil" worry; 5 response categories), and "A3. Body weight" (0: "always much deviation sufficient to cause concern"; 12: "near average at all times"; 4 response categories). Scale B, named "Menstrual pattern (in previous 6 months)" (0: absent menstruations; 12: "regular and cyclic throughout"; 4 response categories). Scale C. "Mental state", is based on what is observed at interview as well as reported abnormalities (0: "grossly abnormal and psychotic with delusions + hallucinations"; 12: "normal"; 4 response categories). Scale D. "Psychosexual state" consists of five subscales: "D1. Attitudes towards sexual matters" (0:"active dislike"; 12: "pleasurable"; 4 response categories), "D2. Professed aims in sexual relationships" (0: "wants to remain single"; 12: "definitely wants to marry and have children or has already done so"; 4 response categories), "D3. Overt sexual behaviour" (0: "avoids heterosexual contacts"; 12: "love affairs with pleasurable sexual relationship (may include married with children)"; 3 response categories), "D4. Attitude to menstruation (if it has returned)" (0: "active dislike"; 12: "pleased that it has returned"; 4 response categories), and "D5. Attitude to menstruation (if it has not returned)" (0: "pleased not returned"; 12: "regrets not returned"; 4 response categories). The D4 and D5 scales were not assessed in male participants. Scale E entitled "Socioeconomic state" is divided into five subscales: "E1. Relationship with nuclear family" (0: "very unsatisfactory"; 12: "satisfactory"; 4 response categories), "E2. Emancipation from family" (0: "many difficulties"; 12: "no difficulties"; 4 response categories), "E3. Personal contacts (apart from family and partner)" (0: "none"; 12: "many close and superficial friends"; 4 response categories), "E4. Social activities" (0: "nil outside family"; 12: "adequate group activities: mixes well outside family"; 4 response categories), and "E5. Employment record" (0: "no paid employment"; 12: "regular full time paid employment without absences"; 4 response categories). An average composite score of all scales (A+B+C+D+E) can be calculated. A separate scale, "Scale G. Self-progress rating", is not included in the composite average score. It is rated by the patient and is scored from 0 to 3, where 0 corresponds to "worse" and 3 to "recovered". In the present study we have concentrated on the responses from each individual on each scale and subscale of the MROAS. The MROAS has been widely used in research and in clinical practice, however, its psychometric properties indicate that it is best suited for AN compared with other EDs [12, 27]. The heteronormative perspective in Scale D has led to the omission of the scale in one study [28]. The measure Morgan Russell general outcome, categorised as good, intermediate or poor, based on weight and menstrual status, has been reported in a previous publication [13].

### Ethics

The Regional Ethical Review Board at the University of Gothenburg approved the study (398–14). All individuals participated voluntarily after giving written informed consent.

## Statistical analysis

This study is a longitudinal study, more specifically a repeated measures analysis. The individual scales and subscales of the MROAS have 3, 4 or 5 response categories, resulting in unbalanced multinomial tables with empty cells. The data structure does not permit the use of statistical methods relying on normality assumptions, consequently, we used exact non-parametric (permutation) statistical methods [29-32]. Computations were performed in the StatXact8 [33] software package on a PC platform. An ad hoc analysis, using Spearman's rank order correlation coefficient [29], was performed to explore a possible correlation between subscale A2 ("Worry about body weight or appearance") and subscale D1 ("Attitudes towards sexual matters") in AN Study 2, AN Study 3, AN Study 4, and AN Study 5, based on a systematic review implying a relationship between body dissatisfaction and sexual dysfunction in EDs [34]. A more detailed description of the statistical procedures is presented as Additional file 1. All tests were two-tailed and conducted at a 5% significance level.

### Results

In the AN group, EDs had successively decreased between AN Study 2 and AN Study 4; 20, 13, and 6 individuals had an ED in AN Study 2, AN Study 3, and AN Study 4, respectively. In AN Study 5, 9 individuals met criteria for an ED (AN: n=3; Binge-eating disorder; BED: n=1; Other Specified Feeding or Eating Disorder (OSFED): n=5). In each follow up study, 3 individuals fulfilled criteria for AN. Only one individual fulfilled the criteria for an ED in the COMP group in AN Study 5, a case of OSFED, night eating syndrome. No one in the COMP group had met criteria for an ED in AN Study 2, AN Study 3, and AN Study 4. Between AN Study 4 and AN Study 5, ten (21.3%) individuals in the AN group

had had an ED relapse (AN: n=5 (in one case, first BED and then AN); BED: n=1 (not including the case that later developed AN); OSFED: n=4). All the 102 individuals (51 AN and 51 COMP), including the 4 dropouts, were alive at 30-year follow-up. Two individuals in the AN group were in treatment for an ED in *AN Study 5*. Twenty-three percent had never received treatment for an ED.

## Overall test AN group versus COMP group

The generalised Cochran–Mantel–Haenszel test was significant for all but two subscales, justifying the detailed follow-up analyses. The two subscales that did not separate well between groups were D4 ("Attitude to menstruation (if it has returned)", test statistic 2.6, exact p = 0.1) and E1 ("Relationship with nuclear family", test statistic 2.79, exact p = 0.1). The generalised Cochran–Mantel–Haenszel test statistic for the remaining scales/ subscales ranged from 8.0 to 56.4 (range of exact p: 0.003 to < 0.00001). Subscale D5 ("Attitude to menstruation (if it has not returned)") could not be analysed as menses had returned in almost all individuals.

Table 1 provides an overview of all the scale and subscale results regarding comparisons between the AN and the COMP group and comparisons within the AN group based on the occurrence of coexistent ASD.

#### Core anorectic symptoms

Subscale A1-Dietary restriction, the distribution across response categories for this variable showed no inter-group difference in AN Study 4 and AN Study 5. Significant inter-group differences were however observed in the first two follow-up studies [10], and were due to a significant improvement in the AN group, but limited to the subgroup 'never ASD' (p = 0.0008). Subscale A2—Worry about body weight or appearance, indicated large or evident differences between the AN and the COMP group in the previous follow-up studies, but the difference between the two groups was not significant in AN Study 5. Subscale A3-Body weight, exhibited a large difference between the AN and the COMP group in AN Study 2 and AN Study 3, but no significant difference in AN Study 4 and AN Study 5. This progress was due to a development towards a more normalised weight in the AN group, and in the COMP group a weight less often near average. The positive development in body weight in the AN group was limited to the group 'never ASD' (p = 0.03).

#### Scale B—Menstrual pattern

There were no between-group differences in *AN Study 5*. The normalised menstrual pattern in the AN group, was only due to the subgroup 'never ASD', which exhibited the positive development (p = 0.0001).

#### Mental state

#### Scale C—Mental State

The between-group differences were highly significant in all four follow-up studies, with many low score responses in the AN group (Table 2). There was no change over time. A statistically significant, negative dose-response relationship was found in all follow-up studies; i.e., the more often an ASD had been assigned, the worse the outcome (Table 3).

#### **Psychosexual state**

Subscale D1—Attitude towards sexual matters, showed a significant between-group difference in all four follow-up studies (*AN Study 2: p*=0.006; *AN Study 3: p*=0.02; *AN Study 4: p*=0.002; *AN Study 5: p*=0.0009). Subscale D2—Professed aims in sexual relationships, had no significant between-group differences in *AN Study 5.* Subscale D3—Overt sexual behaviour, exhibited significant between-group differences in *AN Study 2 (p*=0.0006) and *AN Study 3 (p*=0.04), but no difference in *AN Study 5.* Subscale D4—Attitude to menstruation, if it has returned, indicated that the two groups developed in a similar way, and no significant between-group differences were seen at any follow-up.

#### Socioeconomic state

#### Subscale E1—Relationship with nuclear family

In *AN Study 2* and *AN Study 3* the relationships were reported unsatisfactory in the AN group. Differences between groups were no longer present in *AN Study 4* and *AN Study 5*. No significant dose–response relationships were found regarding ASD in any follow-up.

#### Subscale E2—Emancipation from family

Between-group differences were not significant in AN *Study* 5 (p=0.3). A significant improvement was only seen in the subgroup 'never ASD' (p=0.0008). A significant negative dose–response relationship was visible in *AN Study* 5 (p=0.025); the more often an ASD was assigned, the more difficult the emancipation from family (Table 4).

		A	AN group	vs. COM	P group					١	Vithin Al	N group			
	Omnibus	Group	difference	s at each	follow-	Time	trend	Omnibus	Time	trend withi	n sub-	Dose	-response	effect of A	ASD at
	test <sup>1</sup>		սլ	p <sup>2</sup>		within	groups <sup>3</sup>	test <sup>4</sup>		groups <sup>3</sup>			each fo	ollow-up <sup>3</sup>	
Subscale		AN2 6-year	AN3 10-year	AN4 18-year	AN5 30-year	AN group	COMP group		AN never ASD	AN ASD x1-3	AN ASD x4	AN2 6-year	AN3 10-year	AN4 18-year	AN5 30-year
A1 Dietary	23.5	22.7	11.9	0.4	1.5	+4.1	-2.3	+2.1	+3.4	+1.2	+1.7	+0.9	+1.9	+0.1	+1.2
restriction	<.0001	<.0001	<.001	0.5	0.2	<.0001	0.02	0.03	<.001	0.3	0.1	0.4	0.06	1	0.2
A2 Worry about	43.7	17.8	25.9	5.4	0.8	+2.1	-1.4	-0.7	+1.3	-0.4	+2.4	-0.2	-0.1	+0.4	+0.1
body weight/ appearance	<.0001	<.0001	<.0001	0.02	0.4	0.03	0.2	0.5	0.2	0.7	0.02	0.9	0.9	0.7	0.2
A3 Body weight	15.1	25.5	11.4	0.05	0.04	+2.8	-3.9	-0.8	+2.1	+0.9	+1.4	-0.6	-0.7	+0.2	-0.1
	<.001	<.0001	<.001	0.8	0.8	<.01	<.0001	0.4	0.03	0.4	0.2	0.6	0.6	0.9	1
B Menstrual pattern	8.0	8.1	4.8	0.5	0.1	+3.6	+1.1	-1.4	+3.8	+0.8	+1.1	+0.1	-0.6	-1.9	-0.5
	<.01	<.01	0.03	0.5	0.8	<.001	0.3	0.2	<.001	0.5	0.3	1	0.6	0.05	0.7
C Mental state	56.4	13.7	13.9	15.4	18.7	-1.7	-1.1	-5.1	-0.97	-1.3	-0.6	-2.2	-2.6	-2.8	-2.7
	<.0001	<.001	<.001	<.0001	<.0001	0.1	0.3	<.0001	0.35	0.2	0.6	0.03	0.01	<.01	<.01
D1 Attitude	33.5	7.1	5.3	9.6	11.2	-0.05	+0.3	-4.3	+0.4	-1.4	+0.02	-2.4	-0.7	-3.8	-1.8
towards sexual matters	<.0001	<.01	0.02	<.01	<.001	1	0.8	<.0001	0.7	0.2	1	0.02	0.6	<.001	0.08
D2 Aims in sexual	8.8	2.3	4.7	2.8	0.0	-0.0	-0.9	-3.2	+0.1	+0.8	-0.2	-1.5	-1.4	-1.8	-1.6
relationships	<.01	0.2	0.06	0.09	1	1	0.4	<.01	1	0.6	0.9	0.2	0.2	0.08	0.08
D3 Overt sexual	21.0	11.8	4.5	4.0	2.4	+2.4	+0.6	-5.0	+2.3	+1.2	+1.5	-2.5	-1.6	-4.1	-1.7
behaviour	<.0001	<.001	0.04	0.06	0.1	0.02	0.5	<.0001	0.02	0.3	0.2	0.01	0.1	<.0001	0.1
D4 Attitude to	2.8	0.15	0.9	0.03	0.6	+0.4	-0.1	-1.5	-0.4	+1.2	+0.6	-1.9	-0.8	+0.8	-0.7
menstruation	0.1	0.7	0.3	1	0.5	0.7	0.9	0.1	0.7	0.3	0.6	0.07	0.5	0.6	0.5
E1 Relationship	2.6	2.8	4.5	0.2	1.2	+1.3	-1.0	-0.6	+1.6	-0.3	0	+0.9	-1.1	+0.1	-0.9
with nuclear family	0.1	0.1	0.04	0.6	0.3	0.2	0.3	0.6	0.1	0.8	1	0.4	0.3	1	0.4
E2 Emancipation	10.6	4.5	0.9	7.2	1.2	+2.7	+2.6	-3.3	+3.3	+0.9	-0.2	0.5	-0.5	-3.8	-2.3
from family	<.01	0.03	0.4	<.01	0.3	<.01	0.01	<.01	<.001	0.5	0.9	0.7	0.6	<.001	0.03
E3 Personal	53.6	11.1	23.0	14.6	10.5	-1.0	-1.9	-6.8	-0.65	-1.1	+0.4	-3.5	-2.9	-3.9	-3.1
contacts	<.0001	<.001	<.0001	<.001	<.01	0.3	0.07	<.0001	0.5	0.3	0.7	<.001	<.01	<.0001	<.01
E4 Social activities	48.4	15.8	15.1	9.4	12.0	-1.6	-2.1	-5.1	-1.1	-1.4	-0.3	-2.5	-2.8	-2.8	-2.2
	<.0001	<.0001	<.0001	<.01	<.001	0.1	0.04	<.0001	0.3	0.2	0.9	0.01	<.01	<.01	0.03
E5 Employment	28.9	3.7	8.2	9.9	7.7	-1.2	-0.4	-5.4	-0.6	-1.0	-0.5	-2.0	-3.2	-2.4	-3.4
record	<.0001	0.07	<.01	<.001	<.01	0.3	0.7	<.0001	0.6	0.4	0.6	0.05	<.01	0.01	<.001

**Table 1** Overview of all results on the Morgan Russell outcome assessment schedule scale and subscale regarding comparisonsbetween the AN and the COMP group and comparisons within the AN group based on the occurrence of coexistent ASD

Within each cell, the upper value represents the test statistic, and the lower value represents the exact *p* value. Significant omnibus tests are highlighted in grey. Follow-up tests are highlighted only if the omnibus test was significant. Positive trends are highlighted in blue, negative trends are highlighted in yellow, differences between the AN group and the COMP group are highlighted in green

AN: anorexia nervosa; COMP: comparison; ASD: autism spectrum disorder; AN 2: AN Study 2; six-year follow-up study (AN: n = 51; COMP: n = 51); AN 3: AN Study 3; 10-year follow-up study (AN: n = 51; COMP: n = 51); AN 4: AN Study 4; 18-year follow-up study (AN: n = 51; COMP: n = 51); AN 5: AN Study 3; 10-year follow-up study (AN: n = 51; COMP: n = 51); AN 4: AN Study 4; 18-year follow-up study (AN: n = 51; COMP: n = 51); AN 5: AN Study 5; 30-year follow-up study (AN: n = 47; COMP: n = 51). Year of data collection: AN Study 2: 1991–1992; AN Study 3: 1995–1996; AN Study 4: 2003–2004; AN Study 5: 2015–2016. ASD × 4: Individuals in the AN group with an ASD diagnosis at AN Study 1 to AN Study 4 (all four examinations) (n = 6); ASD × 1–3: those who had been assigned an ASD diagnosis at least once and at most three times (n = 10); never ASD: those who had never fulfilled criteria for ASD (n = 34)

Each statistical analysis was unique, involving a specific 'research question', and type I errors were therefore not taken into account

 $^{\rm 1}$  Generalised Cochran–Mantel–Haenszel test for a singly ordered R  $\times$  C table

 $^2$  Kruskall-Wallis test for a single ordered R  $\times$  C table

 $^3$  Linear-by-Linear association test for a doubly ordered R  $\times$  C table

 $^4$  Generalised Cochran–Mantel–Haenszel test for a doubly ordered R  $\times$  C table

#### Subscale E3—Personal contacts

In all follow-up studies highly significant between-group differences were observed (see Table 5). A significant negative relationship was found in all four follow-up studies according to the detailed dose–response analysis (see Table 6).

#### Subscale E4—Social activities

The AN group had poor results in all follow-up studies, compared with the COMP group (*AN Study* 5: p=0.0005). In all follow-up studies there was a

	Mental state								Singly ordered $\mathbf{R}  imes \mathbf{C}$ (Tables	Contingency
Follow-up	AN group				COMP	group			Kruskal–Wallis test	Inference
	Response cate	egory			Respor	nse category			Test statistic	Exact <i>P</i> value
Carlo	0	4	8	12	0	4	8	12		
6-year	-	10	13	27	0	0	∞	43	13.72	0.00015
10-year	-	10	14	26	-	2	4	44	13.91	0.00016
18-year	2	10	14	25	-	0	7	43	15.43	0.00006
30-year	2	13	15	17	-	2	00	40	18.71	0.00001
Doubly ordered R × C	Linear-by-line: test	ar association	Stratified R × C Contingency Tables Cochran– Mantel– Haenszel Test							
Inference	Test statistic —1.653 Exact <i>p</i> = 0.1	Test statistic —1.147 Exact <i>p</i> < 0.28	Test statistic 56.41 Exact p < 0.00001							
AN: anorexia ne follow-up)	ervosa; COMP: comp	arison; R × C: row	s and columns; 6-year: /	AN study 2 (6-year follow-up);	10-year: AN study 3 (	10-year follow	up); 18-yea	ır: AN study 4 (18	-year follow-up); 30-year: AN study	5 (30-year
Raw data are sit for group diffen	tuated in the middle ences, Kruskall-Walli	e of the table. Resu is test, at each foll	ults of the initial omnibu ow-up are placed to the	s test, a stratified Cochran–M right of the data. Below the c	antel-Haenszel test data the analyses for	for a singly or trendwithin e	dered R × C each group	table, are place (AN or COMP) are	d in the lower right corner. Results e shown (Linear-by-Linear Associat	from the analyses tion test)

$\sim$
$\overline{\mathbf{O}}$
Z
đ
~
2
_~
$\circ$
<u> </u>
ιÔ
č
Ð
£
σ
0
5
_
3
Ó
_
Q
T
σ
0
≒
õ
Ĕ
σ
Δ_
$\geq$
$\overline{}$
2
$\cup$
σ
$\subseteq$
σ
7
2
~
Ç
Ψ
Ψ
5
LT
ŏ
č
5
Š
. <u> </u>
g
얻
6
0
$\cup$
~~~
Ω,
ät
ţ
5
g
Ę
T
¥
$\leq$
$\tilde{\Box}$
$\cup$
Φ
a
Ű
S
_
ß
ŝ
Ą
4
a
ິຫັ
Ĕ,
2
$\geq$
2
-

	Menta	al state														~ -	Analysis for ti trend	me
																	Doubly order C tables	ed R ×
																	Linear-by-Lin Association T	ear est
Dose		AN St	tudy 2		AN 5	tudy 3			ANS	tudy 4			AN	study	6		Fest Statistic	exact p
	Respo	nse cate	gory		Resp	onse c	ategor	~	Resp	onse ca	tegory		Res	oonse	catego	چ ا		
	0	4	8	12	0	4	8	12	0	4	∞	12	0	4	8	12		
Never ASD	0	4	12	18	0	5	6	20	0	9	8	20	0	7	6	14	-0.9656	0.35
ASD × 1–3	0	ŝ	-	9	0		4	Ŋ	0	2	m	5	0	5	2		-1.326	0.2
ASD × 4	<del>, -</del>	m	0	2	-	m	<del>.                                    </del>	-	2	-	m	0	2	-	m		-0.6396	0.6
Dose-response analysis	Linear	-by-line	ar associa	tion test	Stratifie	d R × C	contii	ngency table										
Doubly ordered $\mathbf{R}  imes \mathbf{C}$ tables					Cochrai	i–Mant	el-Hae	enszel test										
Test statistic	-2.165	5 -2.56	52 -2.787	7 -2.695	Test stat	stic -5	119											
Exact <i>p</i>	0.03	0.01	0.006	0.008	Exact <i>p</i> <	0.0000	_											
AN: anorexia nervosa; $R \times C$ : rows diagnosis at AN Study 1 to AN Stuc criteria for ASD (n = 34)	and colur y 4 (all fou	mns; AN s ır examin	study 2: 6-ye ations) (n=	ar follow-up; 6); ASD × 1–;	AN study 3: those w	3: 10-yea no had b	ar follow een ass	/-up; AN study 4: 18 igned an ASD diagr	-year follov rosis at leas	v-up; AN st once ar	study 5: 3 Id at mos	0-year follo t three time	w-up; AS s (n = 10	iD × 4: ); neve	ndividua ASD: th	als in th ose wh	ie AN group wit Io had never ful	th an ASD lfilled
Raw data are situated in the midd for time trend, the Linear-by-Line: Associations test	le of the t ir Associat	able. Resu tion test, ¿	ults of the in are placed to	itial omnibus o the right of	test, a str each leve	atified C	ochran- .' Below	Mantel-Haenszel tu each follow-up stu	est for a do dy we have	ubly orde placed th	ered $\mathbb{R}  imes ($	table, are l of the 'dose	olaced in e-respon	the lov se' ana	ver right yses, usi	corner ng the	. Results of the Linear-by-Linea	analyses ar

Table 3 Morgan-Russell scale C (Mental state); dose-response analysis in the AN group

Dose	Er	nancipat	ion fror	n family	,												Analysis for tin	ne trend
	_																Doubly ordere tables	$d R \times C$
																	Linear-by-linea association tes	ar t
	A	N Study 2	2		AN Stu	ıdy	3		AN	Study	4		AN	Study	5		Test Statistic	Exact P
	Re	esponse o	categor	y I	Respo	nse	categ	ory	Res	ponse	e categ	ory	Res	ponse	e categ	jory		
	0	4	8	12	0	4	8	12	0	4	8	12	0	4	8	12		
Never ASD	2	5	9	19 2	2	2	7	24	0	1	5	29	0	1	3	27	+ 3.269	0.0008
$ASD \times 1-3$	0	2	0	8 (	)	1	1	8	0	1	2	7	0	0	1	9	+0.8723	0.5
$ASD \times 4$	1	2	0	3 (	)	2	1	3	2	1	2	1	1	1	1	3	- 0.1564	0.9
Dose– Response analysis	Linear-	by-linear a	associati	on test	Strat C co table	tified ntin es	d R × gency											
Doubly oredered R × C tables					Coch Man Haei	nran tel– nsze	– I Test											
Test statistic	-0.517	-0.5369	-3.814	-2.314	Test 	stati: 01	stic											
Exact P	0.7	0.6	0.0003	0.025	Exac	t <i>p</i> =	0.001											

Table 4 Morgan-Russell subscale E2 (Emancipation from family); dose-response analysis in the AN group

AN: anorexia nervosa;  $R \times C$ : rows and columns; AN study 2: 6-year follow-up; AN study 3: 10-year follow-up; AN study 4: 18-year follow-up; AN study 5: 30-year follow-up; ASD × 4: Individuals in the AN group with an ASD diagnosis at *AN Study 1* to *AN Study 4* (all four examinations) (n = 6); ASD × 1-3: those who had been assigned an ASD diagnosis at least once and at most three times (n = 10); never ASD: those who had never fulfilled criteria for ASD (n = 34)

Raw data are situated in the middle of the table. Results of the initial omnibus test, a stratified Cochran–Mantel–Haenszel test for a doubly ordered R × C table, are placed in the lower right corner. Results of the analyses for time trend, the Linear-by-Linear Association test, are placed to the right of each level of 'dose'. Below each follow-up study we have placed the results of the 'dose-response' analyses, using the Linear-by-Linear Associations test

significant negative dose–response relationship (AN *Study* 5: p = 0.03).

#### Subscale E5—Employment record

The between-group difference became significant in *AN Study 3* and in the subsequent follow-up studies (*AN Study 3*: p = 0.004; *AN Study 4*: p = 0.0015; *AN Study 5*: p = 0.005). The detailed dose–response analysis showed a clearly significant negative dose–response relationship from *AN Study 3* onwards (see Table 7).

Correlation between Subscale A2 (Worry about body weight or appearance) and Subscale D1 (Attitude towards sexual matters):

The AN group scored significantly worse than the COMP group on the scale Attitude towards sexual matters (D1) from mean age 21 to 44 years. According to a review by Castellini and colleagues [34], sexual dysfunction in women with AN is correlated with greater shape concerns. We therefore performed an ad hoc analysis to investigate whether there was a correlation between Scale A2 (Worry about body weight or appearance) and traits of sexual dysfunction (Scale D1). There were no correlations between Subscale A2 and Subscale D1, neither in the AN (p=0.14), nor in the COMP group (p=1.0) in *AN Study 5*. A correlation was found in the AN group in *AN Study 3* (p=0.02) and *AN Study 4* (p=0.036), and in the COMP group in *AN Study 3* (p=0.007).

### Self-progress rating

## Scale G—Self-progress rating

The probands' self-evaluation of progress was much more positive than the interviewer's evaluations. The test statistics from the generalised Cochran–Mantel–Haenszel test for doubly ordered R × C tables was +0.1491, which gives an exact two-sided p value of 0.9. The test statistics from a linear-by-linear association test was +1.54, which translates into an exact two-sided p value of 0.13. Accordingly, no statistically significant time trends or dose–response relationships were found.

## Discussion

In the present study, a 30-year follow-up of adolescentonset AN, we hypothesised that Mental health, Psychosexual, and Socioeconomic state would still be poorer in the AN group than in the COMP group. In addition, we assumed that comorbid ASD, assessed in *AN Study* 

<b>Table 5</b> Morgan-Russell subscale E3; comparisons betwee	n AN and COMP group at follow-up af	ter 6, 10, 18 and 30 years
---------------------------------------------------------------	-------------------------------------	----------------------------

	Pers	onal contac	ts						Singly ordered R x C co	ntingency tables
	AN	group			COM	P group			Kruskal–Wallis Test	Inference
Follow-up	Res	ponse catego	ory		Resp	onse categ	Jory		Test statistic	Exact P value
	0	4	8	12	0	4	8	12		
6-year	6	11	8	26	0	5	5	41	11.12	0.0009
10-year	4	20	7	20	0	6	1	44	23.02	< 0.00001
18-year	2	13	8	28	1	3	1	46	14.59	0.0001
30-year	4	23	3	17	2	12	1	36	10.47	0.001
Doubly ordered R x C tables	Linear-by-l associatior	inear 1 test	Stratif R x C contir tables	ied Igency						
			Cochr Mante Haens Test	an– el– szel						
Test statistic	-0.9814	-1.867	Test sta 53.63	atistic						
	0.3	0.07	exact <i>p</i> < 0.0	0001						

AN: anorexia nervosa; COMP: comparison; R × C: rows and columns; 6-year: AN study 2 (6-year follow-up); 10-year: AN study 3 (10-year follow-up); 18-year: AN study 4 (18-year follow-up); 30-year: AN study 5 (30-year follow-up)

Raw data are situated in the middle of the table. Results of the initial omnibus test, a stratified Cochran–Mantel–Haenszel test for a singly ordered R × C table, are placed in the lower right corner. Results from the analyses for group differences, Kruskall-Wallis test, at each follow-up are placed to the right of the data. Below the data the analyses for trend within each group (AN or COMP) are shown (Linear-by-Linear Association test)

1 to AN Study 4, would result in poorer outcome evaluated by the MROAS, especially regarding the scales Mental state, Psychosexual, and Socioeconomic state, in the 30-year follow-up. Mental state (C), Attitudes towards sexual matters (D1), Personal contacts (E3), Social activities (E4), and Employment record (E5) all showed significantly worse outcomes in the AN group than in the COMP group at the 30-year follow-up. A dose-response relationship regarding ASD and outcome was observed for the Mental state scale (C), and the socioeconomic scales Personal contacts (E3), Social activities (E4) and Employment record (E5), indicating that the more often an ASD diagnosis had been assigned, the worse the outcome in those specific areas. The results pertaining to the ED-specific scales assessing dieting, menstruation, weight, and preoccupation with body shape and weight were no longer worse in the AN than in the COMP group. However, the improvement in the AN group was limited to those who had never fulfilled the criteria for an ASD.

The AN group reported more problems regarding Attitude towards sexual matters than the COMP group at all follow-up examinations. The results indicate that their libido is not on a par with that of their age-matched controls, despite the majority in the AN group reporting full ED symptom recovery in *AN Study* 5 [13].

Comorbid ASD did not affect the participants' attitudes towards sexual matters anymore. A significant correlation between Attitude towards sexual matters and shape concerns (Worry about body weight or appearance) was seen in *AN Study 3* in the COMP group, and in *AN Study 3* and *AN Study 4*, but not in *AN Study 5*, in the AN group. These findings are, however, difficult to interpret. Our analyses give, at most, limited support for the hypothesis presented by Castellini et al. [34], but the size of our groups limits the statistical power of our analyses. One may surmise that a relationship between excessive concern about body shape and sexual dysfunction is not exclusively an issue for women with an ED, but also for women without an ED.

The favourable results in the present study pertain to the core symptomatology of AN and other EDs, including weight, dieting, body shape and weight concerns, where no significant differences were found in the 30-year follow-up between the AN and the COMP group in these areas. We have previously published anthropometric data from the 30-year follow-up, which show that the mean BMI in the AN group is in accordance with the mean BMI in the COMP group [13]. However, the same publication reported that 19% (n=9) in the AN group still suffered from an ED. Our prospective data have shown that Scale B (Menstrual pattern) and two out of three A

	Pe	rsonal c	ontacts														Analysis for tim	e trend
																	Doubly ordered tables	IR × C
																	Linear-by-linea test	r association
	AN	Study	2		AN	Study 3	3		AN	Study	4		AN	Study 5			Test statistic	Exact P
Dose	Re	sponse	category		Res	ponse	catego	ry	Res	ponse	catego	ry	Res	ponse c	ategory	y		
	0	4	8	12	0	4	8	12	0	4	8	12	0	4	8	12		
Never ASD	2	5	5	22	2	9	6	17	0	6	4	24	0	13	2	15	-0.6536	0.5
$ASD \times 1-3$	2	3	2	3	1	5	1	3	0	5	2	3	2	7	1	0	-1.117	0.3
$ASD \times 4$	2	3	1	0	1	5	0	0	2	2	2	0	2	3	0	1	+0.4708	0.7
Dose– Response analysis	Linear-	by-linea	ır associa	tion test	Stra C co tab	atified ontingo les	R x ency											
Doubly oredered R x C tables					Coo Ma Hae	chran– ntel– enszel 1	test											
Test statistic	-3.462	-2.881	-3.851	-3.095	Test tic-	t statis- —6.798												
Exact P	0.0004	0.004	0.00009	0.0015	exa p<	ct 0.00001												

Table 6 Morgan-Russell subscale E3 (Personal contacts); dose-response analysis in the AN group

AN: anorexia nervosa;  $R \times C$ : rows and columns; AN study 2: 6-year follow-up; AN study 3: 10-year follow-up; AN study 4: 18-year follow-up; AN study 5: 30-year follow-up; ASD × 4: Individuals in the AN group with an ASD diagnosis at *AN Study 1* to *AN Study 4* (all four examinations) (n = 6); ASD × 1-3: those who had been assigned an ASD diagnosis at least once and at most three times (n = 10); never ASD: those who had never fulfilled criteria for ASD (n = 34)

Raw data are situated in the middle of the table. Results of the initial omnibus test, a stratified Cochran–Mantel–Haenszel test for a doubly ordered R × C table, are placed in the lower right corner. Results of the analyses for time trend, the Linear-by-Linear Association test, are placed to the right of each level of 'dose'. Below each follow-up study we have placed the results of the 'dose–response' analyses, using the Linear-by-Linear Associations test

scales (A1 Dietary restriction and A3 Body weight) had already "normalised" by AN Study 4, the 18-year followup, at mean age 32 years [9]. However, in terms of subscale A2 (Worry about body weight or appearance), it was not until the 30-year follow-up that the AN group scored in line with the COMP group. In a prospective follow-up study of female patients with AN, the participants were assessed after nine and 22 years [14]. The definition of recovery focused on not fulfilling any criteria for AN over the last year. Recovery from AN had occurred among 31.4% of the women at the nine-year follow-up, and among 62.8% after 22 years. The authors concluded that recovery from AN "continued over the long term" [14]. In terms of the course of core ED symptoms, we can report similar results and "Worry about body weight or appearance" may be one of the last symptoms to subside before recovery. In a follow-up study of female inpatients with AN twelve and 21 years after their first admission there was no significant improvement in ED symptoms between the two follow-up studies [8, 35]. This differs from the improved outcome of the core ED symptoms between the nine- and 22-year follow-up reported in the study by Eddy's group [14], and a normalisation of core ED symptoms at 18- and 30-year follow-up in the present project.

The Mental health scale showed poorer results in the AN than in the COMP group at all follow-up examinations. The findings are partly due to the expected overrepresentation of EDs in the AN group, which is in line with a previous review of AN outcome studies, where 20% had developed chronic AN [36]. In AN Study 3, 27% still had an ED [21], in AN Study 4 12% had an ED [9], and in the present study, 19% met the criteria for an ED diagnosis [13]. However, over the years, the lower scores on the Mental health scale in the AN group might also be due to the overrepresentation of other psychiatric disorders, in particular, affective and anxiety disorders. The findings of psychiatric morbidity are in line with those of Löwe's group [8], reporting on the outcome of AN 21 years after the first inpatient admission. In addition, Löwe et al. [8], and Halmi et al. [37] observed that alcohol use disorder was a relatively common problem in longterm follow-ups of AN; however, we could not replicate this finding since alcohol use disorder had been rare on all follow-up occasions [9, 13, 21]. In the 30-year followup, we found that the more often a diagnosis of ASD had been assigned, the worse the outcome was regarding

Table 7	Morgan-Russell	l subscale E5	(Employmen	t record); dose–re	sponse ana	lysis in the AN	l grou	р
						/		

	E	Employr	nent re	cord													Analysis for tin	ne trend
	/	AN Stud	y 2		AN	Study	3		AN	Study	4		AN	Study	5		Doubly ordere tables linear-b	d R × C y linear
Dose	F	Respons	e categ	jory	Res	ponse	e cate	gory	Res	ponse	e categ	gory	Res	ponse	e categ	ory	Test statistic	Exact P
	(	) 4	8	12	0	4	8	12	0	4	8	12	0	4	8	12		
Never ASD	3	3 3	5	24	4	1	3	27	8	2	6	19	0	3	11	17	-0.5697	0.6
$ASD \times 1-3$	3	3 1	0	6	0	1	4	5	5	1	0	4	2	2	3	3	-0.9551	0.4
ASD × 4	3	3 0	0	3	4	1	0	1	4	0	1	1	4	0	0	2	-0.528	0.6
Dose– response	Doub	oly ordere	d R x C	tables	Strat cont	ified R ingenc	× y											
analysis	Linea	r-by-line	ar assoc	iationtes	t Coch Man Haei	nran– tel– nszel												
Test statistic	-2	-3.22	-2.43	7 —3.39	Test stic—	statis- 5.427												
Exact p	0.05	0.0015	0.014	0.0007	Exact p < 0.	t .00001												

AN: anorexia nervosa;  $R \times C$ : rows and columns; AN study 2: 6-year follow-up; AN study 3: 10-year follow-up; AN study 4: 18-year follow-up; AN study 5: 30-year follow-up; ASD × 4: Individuals in the AN group with an ASD diagnosis at *AN Study 1* to *AN Study 4* (all four examinations) (n = 6); ASD × 1-3: those who had been assigned an ASD diagnosis at least once and at most three times (n = 10); never ASD: those who had never fulfilled criteria for ASD (n = 34)

Raw data are situated in the middle of the table. Results of the initial omnibus test, a stratified Cochran–Mantel–Haenszel test for a doubly ordered R × C table, are placed in the lower right corner. Results of the analyses for time trend, the Linear-by-Linear Association test, are placed to the right of each level of 'dose'. Below each follow-up study we have placed the results of the 'dose–response' analyses, using the Linear-by-Linear Associations test

the Mental health scale. This finding was also true with regard to *AN Study 2, AN Study 3* and *AN Study 4,* which probably reflects the pervasive impact an ASD has on an individual's mental health.

Emancipation from family showed no improvement between the ten-year (AN Study 3) and the 30-year follow-up (AN study 5) in the subgroup of individuals in the AN group that had been assigned an ASD diagnosis on at least one occasion. In a Swedish register study, where female inpatients with adolescent-onset AN were followed up 9-14 years after admission, the women were significantly more often still living with their parents than age- and sex-matched individuals in the general population (9.0% versus 6.8%) [38]. One can surmise that ASD may have been overrepresented among the former AN inpatients in the register study, since ASD is more common among individuals with severe and enduring AN [15]. In any case, with an adult with AN who has not become independent of his or her parents, ASD should therefore be considered.

The socioeconomic subscales; i.e., Personal contacts, Social activities and Employment record, were still worse in the AN group compared with the COMP group at the 30-year follow-up. In the AN group, the presence of ASD contributed to the poorer outcome on these subscales. A study comparing social activities among women and men with ASD without intellectual disability found that women more often than men stated that they preferred their own company [39]. Our findings regarding less interest in taking part in social interplay are in line with the core symptomatology of ASD; however, a long-lasting ED together with another psychiatric morbidity probably also contribute to a limited social sphere.

'Employment record' had been worse in the AN than in the COMP group since AN Study 3. The more often an ASD diagnosis had been assigned, the worse the employment record during the last 20 years. Even poorer results in terms of employment record were reported in a 20-year follow-up study of AN patients from the 1990s, where half of the participants were unemployed [7]. Our discouraging Employment record results may have several explanations. Firstly, the information may reflect that some of those with ASD who had previously reported that they were engaged in university studies had later not been able to apply their skills in a workplace. Secondly, most individuals in the AN group no longer had small children and the individuals with ASD, who previously had been on full-time parental leave, now had difficulty finding suitable work, due to their social interaction impairment. Both examples indicate how individuals with previous or current AN might try to camouflage their poor psychosocial functioning from society's expectations when it comes to coping with a full-time job.

#### Strengths and limitations

This study has many strengths. We have followed a group of individuals with adolescent-onset AN for almost three decades. The mean age at AN onset was 14 years (30 years before the present study), and the first examination was performed at mean age 16 years. The study has a prospective case-control design; i.e., both the AN and the COMP group were given identical in-depth test batteries. All individuals were recruited from the community, and half the AN group constituted a total age cohort of adolescents attending 8th grade at school in Gothenburg in 1985. No other AN sample has been followed up on so many occasions, and over such a long period of time. In our previous studies we have never experienced any dropout, and at the current assessment, 96% of the total sample participated, with four of the individuals in the AN group abstaining. Our research group's early recognition of comorbid ASD in AN enabled us to assess ASD traits beginning already in the original study in 1985.

Some limitations should also be considered. Firstly, ASD was not assessed in the present study, and for this reason, we cannot know for certain whether the individuals with a previous ASD still fulfil the diagnostic criteria. We can, however, surmise that the diagnostic stability regarding the 'ASD  $\times$  4' group is fairly robust. Secondly, the sample consisting of 51 AN cases and 51 COMP cases may be considered relatively small. The modest sample size has, however, enabled us to conduct in-depth face-to face interviews at all examinations, and has made it possible to follow up every participant at each assessment, except four in the current study. The sample size limited the statistical power. Thirdly, this was not a treatment outcome study and therefore interventions were not systematically evaluated. Twentythree per cent in the AN group reported that they had never received any treatment for an ED. However, the outcomes according to the Global Assessment of Functioning (GAF) [19] and the Morgan Russel averaged scale score did not differ between those who had ever received treatment for an ED and those who had not received treatment [13].

## Conclusions

Thirty years after the onset of AN, mental health, libido, social interaction and employment were still not on a par with the general population. The presence of ASD, diagnosed on one or several occasions, seemed to be one major reason why a minority of individuals with adolescent-onset AN never attained an acceptable level of psychosocial functioning. Among those with no history of ASD, the core ED symptomatology had normalised after three decades. In individuals who do not respond to evidence-based treatment of AN, screening procedures for ASD must be considered at an early stage.

#### Abbreviations

AN: Anorexia nervosa; ASD: Autism spectrum disorder; COMP: Comparison; ED: Eating disorder; HRQoL: Health-Related Quality of Life; MROAS: Morgan Russell Assessment Schedule.

## Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s40337-021-00518-1.

Additional file 1: Data structure and Statistical analyses

#### Acknowledgements

The authors gratefully acknowledge the patients who participated in the original study and the follow-ups. Birgitta Melander ensured the coverage of the original cohort of participants.

#### Authors' contributions

SN designed the study, conducted the statistical analyses, analysed and interpreted the data, drafted the manuscript and critically revised the manuscript for important intellectual content. SD, LD, and CaG participated in the acquisition of the data and critically revised the manuscript for important intellectual content. CG and MR designed the study and critically revised the manuscript for important intellectual content. EW designed the study, participated in the acquisition of the data, analysed and interpreted the data, drafted the manuscript and critically revised the manuscript for important intellectual content. All authors read and approved the final manuscript.

#### Funding

Open access funding provided by University of Gothenburg. Dr. Wentz received support from the Jane and Dan Olsson Foundations (2015 and 2016-55), the Wilhelm and Martina Lundgren Foundation (vet2-73/2014, and 2017-1555), the Petter Silfverskiöld Memorial Foundation (2016-007), and Swedish State Support for Clinical Research (#ALFGBG-813401). Sandra Rydberg Dobrescu received support from the Royal and Hvitfeldt Foundation (2016), and the Foundation for Queen Silvia Children's Hospital (2018). Lisa Dinkler was supported by Queen Silvia's Jubilee Fund (2016) and the Samariten Foundation (2016-0150). Dr. Christopher Gillberg received grant support from the Swedish Research Council (521-2012-1754), the AnnMari and Per Ahlqvist Foundation, and Swedish State Support from the Birgit and Sten A. Olsson Foundation for research into mental disabilities.

#### Availability of data and materials

The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

#### Declarations

#### Ethics approval and consent to participate

The Regional Ethical Review Board at the University of Gothenburg approved the study (398–14). All individuals participated voluntarily after giving written informed consent.

#### **Consent for publication**

Not applicable.

#### **Competing interests**

The authors declare that they have no conflict of interest.

#### Author details

<sup>1</sup>Psychiatric Research Unit, Psychiatry Region Zealand, Slagelse, Denmark. <sup>2</sup>Gillberg Neuropsychiatry Centre, Institute of Neuroscience and Physiology, University of Gothenburg, Gothenburg, Sweden. <sup>3</sup>Department of Child and Adolescent Psychiatry, University of Glasgow, Glasgow, UK. <sup>4</sup>Department of Clinical Sciences Lund, Child and Adolescent Psychiatry, Lund University, Lund, Sweden. <sup>5</sup>Department of Psychiatry and Neurochemistry, Institute of Neuroscience and Physiology, University of Gothenburg, Gothenburg, Sweden. <sup>6</sup>Högsbo Hospital, Lilla Kapplandsgatan 26, 42137 Västra Frölunda, Sweden.

#### Received: 29 April 2021 Accepted: 30 November 2021 Published online: 09 January 2022

#### References

- Arcelus J, Mitchell AJ, Wales J, Nielsen S. Mortality rates in patients with anorexia nervosa and other eating disorders: a meta-analysis of 36 studies. Arch Gen Psychiatry. 2011;68(7):724–31.
- Fayssoil A, Melchior JC, Hanachi M. Heart and anorexia nervosa. Heart Fail Rev. 2021;26(1):65–70.
- 3. Stoving RK. Mechanisms in endocrinology: anorexia nervosa and endocrinology: a clinical update. Eur J Endocrinol. 2019;180(1):R9–27.
- Kornreich L, Shapira A, Horev G, Danziger Y, Tyano S, Mimouni M. CT and MR evaluation of the brain in patients with anorexia nervosa. AJNR Am J Neuroradiol. 1991;12(6):1213–6.
- Fichter MM, Quadflieg N. Mortality in eating disorders results of a large prospective clinical longitudinal study. Int J Eat Disord. 2016;49(4):391–401.
- Rosling AM, Sparen P, Norring C, von Knorring AL. Mortality of eating disorders: a follow-up study of treatment in a specialist unit 1974–2000. Int J Eat Disord. 2011;44(4):304–10.
- Ratnasuriya RH, Eisler I, Szmukler GI, Russell GF. Anorexia nervosa: outcome and prognostic factors after 20 years. Br J Psychiatry. 1991;158:495–502.
- Lowe B, Zipfel S, Buchholz C, Dupont Y, Reas DL, Herzog W. Long-term outcome of anorexia nervosa in a prospective 21-year follow-up study. Psychol Med. 2001;31(5):881–90.
- Wentz E, Gillberg IC, Anckarsater H, Gillberg C, Rastam M. Adolescent-onset anorexia nervosa: 18-year outcome. Br J Psychiatry. 2009;194(2):168–74.
- 10. Nielsen S, Anckarsater H, Gillberg C, Gillberg C, Rastam M, Wentz E. Effects of autism spectrum disorders on outcome in teenage-onset anorexia nervosa evaluated by the Morgan-Russell outcome assessment schedule: a controlled community-based study. Mol Autism. 2015;6:14.
- Fichter MM, Quadflieg N, Crosby RD, Koch S. Long-term outcome of anorexia nervosa: results from a large clinical longitudinal study. Int J Eat Disord. 2017;50(9):1018–30.
- Morgan HG, Hayward AE. Clinical assessment of anorexia nervosa. The Morgan-Russell outcome assessment schedule. Br J Psychiatry. 1988;152:367–71.
- Rydberg Dobrescu S, Dinkler L, Gillberg IC, Råstam M, Gillberg C, Wentz E. Anorexia nervosa: 30-year outcome. Br J Psychiatry. 2020;216:97–104.
- Eddy KT, Tabri N, Thomas JJ, Murray HB, Keshaviah A, Hastings E, et al. Recovery from anorexia nervosa and bulimia nervosa at 22-year followup. J Clin Psychiatry. 2017;78(2):184–9.
- Wentz E, Lacey JH, Waller G, Rastam M, Turk J, Gillberg C. Childhood onset neuropsychiatric disorders in adult eating disorder patients: a pilot study. Eur Child Adolesc Psychiatry. 2005;14(8):431–7.
- Rastam M, Gillberg C, Garton M. Anorexia nervosa in a Swedish urban region: a population-based study. Br J Psychiatry. 1989;155:642–6.
- Rastam M. Anorexia nervosa in 51 Swedish adolescents: premorbid problems and comorbidity. J Am Acad Child Adolesc Psychiatry. 1992;31(5):819–29.
- APA. Diagnostic and statistical manual of mental disorders, of mental disorders, 3rd edition revised (DSM-III-R). Washington DC: American Psychiatric Press; 1987.
- 19. APA. Diagnostic and statistical manual of mental disorders, 4th edition (DSM-IV). Washington DC: American Psychiatric Press; 1994.

- Gillberg IC, Rastam M, Gillberg C. Anorexia nervosa outcome: six-year controlled longitudinal study of 51 cases including a population cohort. J Am Acad Child Adolesc Psychiatry. 1994;33(5):729–39.
- Wentz E, Gillberg C, Gillberg IC, Rastam M. Ten-year follow-up of adolescent-onset anorexia nervosa: psychiatric disorders and overall functioning scales. J Child Psychol Psychiatry. 2001;42(5):613–22.
- Dewey M. Living with Asperger's syndrome. In: Frith U, editor. Autism and Asperger syndrome. Cambridge: Cambridge University Press; 1991. p. 184–206.
- Baron-Cohen S, Wheelwright S, Skinner R, Martin J, Clubley E. The autism-spectrum quotient (AQ): evidence from Asperger syndrome/highfunctioning autism, males and females, scientists and mathematicians. J Autism Dev Disord. 2001;31(1):5–17.
- Sheehan DV, Lecrubier Y, Sheehan KH, Amorim P, Janavs J, Weiller E, et al. The Mini-International Neuropsychiatric Interview (M.I.N.I.): the development and validation of a structured diagnostic psychiatric interview for DSM-IV and ICD-10. J Clin Psychiatry. 1998;59 Suppl 20:22–33;quiz 4–57.
- First M, Gibbon M, Williams J. Structured clinical interview for DSM-IV Axis I Disorders - Patient edition (SCID-I/P, Version 2.0, 4/97 revision). New York: Biometrics Research Department; 1997.
- Morgan HG, Russell GF. Value of family background and clinical features as predictors of long-term outcome in anorexia nervosa: four-year followup study of 41 patients. Psychol Med. 1975;5(4):355–71.
- Lund C, Jorgensen J, Stage KB, Sorensen T. Interrater reliability of a Danish version of the Morgan Russell scale for assessment of anorexia nervosa. Int J Eat Disord. 1999;25(1):105–8.
- Lange CRA, Ekedahl Fjertorp H, Holmer R, Wijk E, Wallin U. Long-term follow-up study of low-weight avoidant restrictive food intake disorder compared with childhood-onset anorexia nervosa: psychiatric and occupational outcome in 56 patients. Int J Eat Disord. 2019;52(4):435–8.
- Siegel S, Castellan NJ. Nonparametric statistics for the behavioral sciences. 2nd ed. New York: McGraw-Hill; 1988.
- 30. Agresti A. Categorical data analysis. 2nd ed. New York: Wiley; 2002.
- Mantel N, Byar DP. Marginal homogeneity, symmetry and independence. Communication in Statistics. 1978; series A7:953–76.
- 32. Landis R, Heyman ER, Koch GG. Average partial association in three-way contingency tables: a review and discussion of alternative tests. Int Stat Rev. 1978;46:237–54.
- 33. Cytel I. StatXact8 User Manual. Cambridge: Cytel Inc.; 2007.
- Castellini G, Lelli L, Ricca V, Maggi M. Sexuality in eating disorders patients: etiological factors, sexual dysfunction and identity issues: a systematic review. Horm Mol Biol Clin Investig. 2016;25(2):71–90.
- Deter HC, Herzog W. Anorexia nervosa in a long-term perspective: results of the Heidelberg-Mannheim Study. Psychosom Med. 1994;56(1):20–7.
- Steinhausen HC. Outcome of eating disorders. Child Adolesc Psychiatr Clin N Am. 2009;18(1):225–42.
- Halmi KA, Eckert E, Marchi P, Sampugnaro V, Apple R, Cohen J. Comorbidity of psychiatric diagnoses in anorexia nervosa. Arch Gen Psychiatry. 1991;48(8):712–8.
- Hjern A, Lindberg L, Lindblad F. Outcome and prognostic factors for adolescent female in-patients with anorexia nervosa: 9- to 14-year follow-up. Br J Psychiatry. 2006;189:428–32.
- Baldwin S, Costley D. The experiences and needs of female adults with high-functioning autism spectrum disorder. Autism. 2016;20(4):483–95.

## Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.